MA194 { Using Unix

Instructor: Dan Chaney (chaney@ms.uky.edu) Office hours: Mon 2-3, 606-257-2150

Class #26 - Tuesday, April 16, 1996 – Week 13

1996 NCAA Men’s Basketball Champion: UK Wildcats! Community Colleges: Tues 9:00-9:50 & 10:00-10:50 PM EST UK/Lexington: Wed & Fri 9:00 PM EST, TCI Channel 16

URL: http://www.ms.uky.edu/~ma194

Mailing list: ma194-class@ms.uky.edu To join the mailing list, send a note to ma194@ms.uky.edu

Tonight’s Topics ffl Homework, missed tapes and the IRS ffl A shell help system ffl Making forms in HTML ffl Converting a shell script to CGI

Slides for this class are the same as the previous class